Author's response to reviews

Title: Stent versus gastrojejunostomy for the palliation of gastric outlet obstruction: a systematic review

Authors:

Suzanne M Jeurnink (s.jeurnink@erasusmc.nl)
Casper HJ van Eijck (c.vaneijck@erasusmc.nl)
Ewout W Steyerberg (e.steyerberg@erasusmc.nl)
Ernst J Kuipers (e.j.kuipers@erasusmc.nl)
Peter D Siersema (p.siersema@erasusmc.nl)

Version: 3 Date: 29 May 2007

Author's response to reviews: see over
Dear Editorial Production Team,

Thank you for reviewing the format of the manuscript, entitled:

"Stent versus gastrojejunostomy for the palliation of gastric outlet obstruction: a systematic review."

Point by point we will discuss the comments you had:

1. The list of authors in the manuscript should be written exactly as they are in the submission system.

   Corrected.

2. Remove line numbers.

   Line numbers have been removed.

3. Please use sentence case for all headings.

   Corrected.

4. Please rename the section ‘Methods’.

   Corrected.

5. Please remove any highlighting.

   Highlighting have been removed from the manuscript.

6. Author’s contribution -> Authors’ contribution

   Corrected.

7. Please acknowledge anyone who contributed towards the study.

   We do not have any acknowledgements to make.

8. Fifty-four references appear in the manuscript actual and fifty-three within the submission system. Please check.

   The references have been checked. One reference was mentioned double in the manuscript, which was corrected.

9. Tables as additional files
Tables have been inserted in the manuscript file. Corresponding additional files have been deleted from the submission system.

10. Tables

Tables are divided into cells/fields. The borders are visible black lines. The tables do not contain tabbed text. Text alignment is clear and consistent.

We hope that we have answered the raised issues and comments correctly and we are very pleased that BMC Gastroenterology will publish our manuscript.

For any more questions and/or comments, you are more than welcome to contact us.

Sincerely yours,
on behalf of all co-authors,

Suzanne M. Jeurnink Peter D. Siersema